Author's response to reviews

Title: Endothelial nitric oxide synthase gene polymorphism (Glu298Asp) and development of pre-eclampsia: a nested case-control study and a meta-analysis

Authors:

Christina KH Yu (chrissiyu@aol.com)
Juan P Casas (rmhajpc@ucl.ac.uk)
Makrina D Savvidou (msavvidou2005@yahoo.co.uk)
Manpreet K Sahemey (manpreetsahemey@yahoo.com)
Kypros H Nicolaides (kypros@technocom.com)
Aroon D Hingorani (a.hingorani@ucl.ac.uk)

Version: 6 Date: 22 February 2006

Author's response to reviews: see over
Dear Editor,

I enclosed the final version of the revised manuscript with all the formatting changes as requested.

Thank you

Christina Yu
Harris Birthright Research Centre for Fetal Medicine,
King’s College Hospital,
Denmark Hill,
London SE5 9RS, U.K.
Tel : 0044 207 346 3070
Email chrissieyu@aol.com